Respond to reviewers' comments

1. We found that the language editing company mentioned in your submitted language certificate only polished the initial manuscript. As there are many changes in the content of your manuscript during the revision process, there are still some language problems in the revised manuscript, and it still requires further language polishing to fix grammatical and other errors to meet the publication requirement (Grade A). Now, you are requested to send the revised manuscript to a professional English language editing company or a native English speaking expert to polish the manuscript further. When you submit the polished manuscript to us, you must provide a new language certificate along with the manuscript. Once you finish this, your manuscript will be quickly accepted and published online. Please visit the following website for the professional English language editing companies we recommend: https://www.wjgnet.com/bpg/gerinfo/240.

Answer: Thanks to finding the problem. Manuscripts have been proofread by native English speaking experts and up-to-date language editing certificates are available. NO. 545B-8ED7-2734-F90E-45DF.

2. Please provide a running title (no more than six word).

Answer: Thank you for the suggestions, which are very helpful for us to improve my manuscript. Running title: Based on TCGA cohort analysis.

3. The author order of copyright licence agreement did not match the author order in manuscript.

Answer: Thank you for finding this error. All authors have re-signed the copyright licence agreement and uploaded.

4. The crosscheck report the manuscript was similar to a preprint https://www.researchsquare.com/article/rs-127782/v1, make sure the article had not been formally published in all or part of other journals before.

Answer: Thank you for your confirmation. We are confident that the manuscript has not been formally published in all or part of other journals. All authors are guaranteed.